Author's response to reviews

Title: Phellodendron and Citrus extracts benefit cardiovascular health in osteoarthritis patients: a double-blind, placebo-controlled pilot study

Authors:

Julius Oben PhD (juliusoben@hotmail.com)
Ebangha Enonchong MSc (juliusoben@hotmail.com)
Shil Kothari MS (shilkothari@hotmail.com)
Walter Chambliss PhD (chamblissw@aol.com)
Robert Garrison RPh (ddolnick@nextpharmaceuticals.com)
Deanne Dolnick MS (ddolnick@nextpharmaceuticals.com)

Version: 3 Date: 16 May 2008

Author's response to reviews: see over
Dear Nutrition Journal,

Thank you for your e-mail dated May 15, 2008. We are replying to each of your requested formatting changes as best we understand them.

1. **Author list** - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.
   
The only differences we could find between the authors list in the submission system and that in the manuscript were in the e-mail addresses. We have added an e-mail address for EE in the manuscript. There is no e-mail for RG, as he is deceased. In the submission system we used the e-mail for DD, but we felt that was not necessary for the manuscript as it clearly states that he is deceased.

2. **Please can you remove the additional files.**
   
   Done

3. **Tables - Please note that we are unable to display vertical rules within the tables nor are we able to write text vertically within the cells.**
   
   We have reformatted the tables so that each row is separated. We hope this is what you requested.

4. **please make reference to table 6 within the manuscript.**
   
   Done

If for any reason we have misunderstood your requests, please let us know. We have endeavored to comply.
Sincerely,
Deanne Dulnick